Author's response to reviews

Title: Screening of mental health and substance users in frequent users of a general Swiss emergency department

Authors:

Francis Vu (francis.vu@chuv.ch)
Jean-Bernard Daeppen (jean-bernard.daeppen@chuv.ch)
Olivier Hugli (olivier.hugli@chuv.ch)
Katia Iglesias (katia.iglesias@unine.ch)
Stephanie Stucki (sstucki@addictionsuisse.ch)
Sophie Paroz (sophie.paroz@chuv.ch)
Marina Canepa Allen (marina.canepa@hotmail.com)
Patrick Bodenmann (patrick.bodenmann@hospvd.ch)

Version: 3 Date: 23 September 2015

Author's response to reviews: see over
Dear Editor and Reviewers,

Thank you for having reviewed thoroughly our manuscript entitled: *Screening of mental health and substance users in frequent users of a general Swiss emergency department.*

Once again, we are very grateful for the interesting and useful questions, comments and suggestions made by the reviewers and the editor that helped improve the quality of our manuscript.

We revised the main document with the following editorial revisions requests:

1. Include the name of the ethics committee

   The name of the ethics committee was inserted in the text (lines 83-84 of the revised document), as following: “This research was approved by the Ethics Committee of the University of Lausanne (reference number 156/09)”

2. Consent

   Corrections were made (lines 84-85 of the revised document), as following: “We obtained written consent from all study participants.”
Yours sincerely,

Patrick Bodenmann
MD, MSc PH, Senior lecturer,
Senior Faculty

Francis Vu
Senior resident, MD candidate